Author’s response to reviews

Title: Generic Quality of Life in Persons with Hearing Loss: A Systematic Literature Review

Authors:

Øyvind Nordvik (oyno@hvl.no)
peder Heggdal (pederheggdal@gmail.com)
Jonas Brännström (jonas.brannstrom@med.lu.se)
Flemming Vassbotn (flemming.slinning.vassbotn@helse-bergen.no)
Anne Kari Aarstad (anne.k.aarstad@uis.no)
Hans Jørgen Aarstad (hans.jorgen.aarstad@helse-bergen.no)

Version: 1 Date: 07 Apr 2017

Author’s response to reviews:

Dear editors.

We submit the manuscript “Generic Quality of Life in Persons with Hearing Loss: A Literature Review” for consideration for publication in BMC Ear, Nose, and Throat Disorders.

Please note that abbreviations and declarations section have been moved to after conclusions (page 19), according to editors feedback.

Sincerely, Øyvind Nordvik